Reviewer #1:

**Scientific Quality:** Grade B (Very good)

**Language Quality:** Grade A (Priority publishing)

**Conclusion:** Accept (General priority)

**Specific Comments to Authors:** The manuscript adequately addresses the subject and highlights an unusual complication of using the nasojejunal catheter for feeding. However, the described complication can turn into a disaster, adding morbidity and mortality to patients. And it adds an economical and also adequate solution as We introduced a way of ultrasound-guided jejunum feeding tube placement to avert the disaster, which was convenient and economical like ultrasound guided jejunal feeding probe placement.

**Response to comment:** Thank you for your valuable suggestions and evaluation of our team’s article.